Author's response to reviews

Title: Short-term mortality after perforated or bleeding peptic ulcer among elderly patients: a population-based cohort study

Authors:

Steffen Christensen (sc@dce.au.dk)
Anders Riis (ar@dce.au.dk)
Mette Norgard (m.noergaard@rn.dk)
Henrik T Sorensen (hts@dce.au.dk)
Reimar W Thomsen (r.thomsen@rn.dk)

Version: 5 Date: 17 April 2007

Author's response to reviews:

Please remove the email addresses in the author list.

Please put the acknowledgements after the authors' contributions.

Dear Editor
We have reformatted the manuscript according to your suggestions and hope the changes have made the manuscript ready for publication.
Please remove the email addresses in the author list.

Please put the acknowledgements after the authors' contributions.

Please rename the figure titles in the figure document to a and b. The titles should go in the figure legend.

Please give the figure section a heading: 'Figure legends', and the tables also i.e. 'Tables'.

Please rename the figure titles in the figure document to a and b. The titles should go in the figure legend.

Please give the figure section a heading: 'Figure legends', and the tables also i.e. 'Tables'.

Sincerely
Steffen Christensen